

http://dx.doi.org/10.12785/ijcds/100135

# Management of WSN-enabled Cloud Internet of Things: A Review

## Edje Efetobor Abel<sup>1</sup> and Abd Latiff Shafie Muhammad<sup>2</sup>

<sup>1,2</sup> Department of Computer Science, Universiti Teknologi Malaysia (UTM), Johor Bahru, Malaysia <sup>1</sup>Department of Computer Science, Delta State University (DELSU), Abraka, Nigeria

Received 2 Mar. 2020, Revised 30 Apr. 2020, Accepted 3 Aug. 2020, Published 8 Feb. 2021

**Abstract:** Internet of Things (IoT) is the networking of computer devices and objects to share information through the internet. The last seven years have witnessed great success in deploying massive cloud storage and computation resources to leverage IoT devices' limitations. However, IoT sensing device(s) such as Wireless Sensor Network (WSN) is still challenged with inefficient sensing data provisioning in real-time basis. Therefore, this paper investigates the existing algorithms developed by previous research to manage WSN to retrieve efficient and desired sensed data on the Cloud IoT paradigm. The background information is presented about the origin of WSN-enabled Cloud IoT and its current developmental stride. We identify and analyze how the algorithms can manage the WSNs to resolve the challenges mentioned above. Furthermore, we discuss how the algorithms are evaluated and validated by useful Quality of Service (QoS) delivery metric parameters. As stated in previous research, the future performance improvement of the algorithms, which pave the for future research directions, is also presented in this paper and ends with a concluding remark.

Keywords: Wireless Sensor Network, Internet of Things, Cloud Data Center, Quality of Service Deliver Parameters

## 1. INTRODUCTION

The integration of Cloud Computing and the Internet of Things (IoT) has revolutionized the deployment and usage of Information Communication Technology in recent times. Presently, it is used to provide on-demand sensory data resource(s) and application service in various fields such as healthcare, manufacturing industries, education, agriculture, and smart city. IoT is a network of interconnected things or device(s) which are embedded with sensors, software, network connectivity and related electronics for the retrieval and exchange of data from physical objects (e.g., environment, fridges, and vehicles) [1]. It is a framework that integrates the physical world and computer systems, enabling both to communicate and share information over an existing network (e.g., Internet) connectivity. For instance, a lightweight platform can be deployed in an individual's home to collect data from connected objects such as the fridge, electric lights, and heating systems [2]. The data obtained are transmitted through a private or public network to a computer server and other accessible storage devices. IoT device(s) generates a massive amount of data with various connotations and limited storage space, processing

capability, and energy power source. On the other hand, the cloud massive storage resource and data-intensive computational tasks (e.g., data fusion and coordination capabilities) for achieving more excellent data quality and energy efficiency. Cloud computing is a service delivery model of dynamic, scalable and virtualized resources as through the Internet [3]. Therefore, integrating both Cloud and IoT can support IoT devices' requirements, which leads to the provisioning of new and exciting data resource(s) and applications to the end-users [4]. This is because the features of the cloud are utilized to resolve the limitations of IoT devices. The Cloud-enabled IoT infrastructure's immediate challenges are the massive data generated from heterogeneous, dynamic and seamless IoT devices [5]. IoT device(s) such as Wireless Sensor Networks (WSNs) are powered by battery energy that depletes within a limited operation period. These operations include sensing coverage, radio communication, data processing, and sampling. Therefore, there is a great need to conserve sensor nodes' energy consumption in the WSNs during operation. Previous research surveys on Cloud-IoT infrastructure mainly focus on a particular research domain such as the survey on applications and challenges of Cloud-IoT, an overview of sensor cloud infrastructure and a review on

*E-mail: eaedje@graduate.utm.my, shafie@utm.my* 



security management of Docker cluster in the cloud. Therefore, this research aims to explore and analyze the deployment of existing algorithms developed to manage WSNs for efficient provisioning of desired data on Cloud IoT infrastructure. Our contributions in this research are as follows;

- We present a piece of background information about the WSNs-enabled Cloud IoT's main components, its origin, and current developmental strides.
- We identify and analyze how the existing algorithms can manage the physical features of WSNs for efficient provisioning of desired sensing data on the Cloud IoT platform.
- We discuss the efficient quality of service delivery (QoS) metric parameters used to evaluate and validate the existing algorithms' performance.
- We highlight the challenges resolved, benefits and the limitations of existing algorithms that will motivate further research in this field in the foreseeable future.

The rest of this paper is structured as follows; Section II presents the background information of WSN enabled Cloud IoT. Section III discusses the research methodology adopted to actualize the current research survey understudy, while Section IV presents brief related research works in Cloud IoT. Section V comprises a detailed analysis of the existing algorithms, highlighting the issues they resolved, results (benefits) and future improvements in a tabular form. Section VI discusses the efficient QoS delivery metric parameters used to evaluate and validate the existing algorithms. Section VII contains discussion and concluding remarks.

## 2. BACKGROUND INFORMATION

Cloud and the Internet of Things are two different technologies that are presently making a new trend in the computing industry. The phrase "Internet of Things" was invented some two decayed ago by the founders of the MIT Auto-ID Center, co-headed by Kevin Ashton in 1999 and David L. Brock in 2001. Auto-ID is defined as any broad class of identification technologies used to automate, minimize errors, and improve efficiency [6]. Devices such as smart cards, barcodes, sensors, biometrics, and voice recognition are classified as Auto-ID technologies. However, Radio Frequency Identification (RFID) has been the dominant Auto-ID technology in use since 2005. RFID is a technology that enables computers to identify and track objects (e.g., raw materials and finished products) automatically as they flow from the production stage to the distribution center and ends in the shelves of shop outlets. It is mostly used

in manufacturing industries and Courier services for identification and monitoring of goods on transits in real time basis. In that same year, Kevin Ashton stated that "IoT is considered as the extension of Radio Frequency Identification, in the form of amoeba in wireless computing world" [6]. IoT was formally introduced in 2007 when the International Telecommunications Union initially circulated the report on the subject. The Union categorically adopts a holistic method suggesting that IoT interconnect things (e.g., sensor devices) intelligently by labeling and curbing things using Wireless Sensor Networks (WSNs) and Embedded Systems. It is envisaged that in the year 2020, the development and adoption of WSN and smart sensor devices will increase to 50 billion on-demand. The primary function of WSNs is to capture and divulge live events, which are converted into data format can be processed and stored [7]. Consequently, cloud computing is an information technology business model delivered as a service through the Internet. Both hardware and software computing services are dispatched on-demand basis to end-users in a self-service manner with minimum service provider interference [8]. Cloud computing came into the limelight in the year 1959 by John Macarthy. The ideology of developing a time-shared computers that can allow the sharing of computing resources (e.g., applications and hardware) between two or more end-users via multitasking instigated the evolution of cloud computing. He further stated that computers should become a utility like the telephone services in the year 1961, which motivated Douglas Parkhill to base his research on utility computing's feasibility in the year 1966. His research results depict that the tendency for the actualization of utility computing services is not feasible. In 2004, Salesforce.com began to provide cloud services such as Software as a Service (SaaS) on the World Wide Web to end-users on a "pay-as-you-use" basis. Consequently, Google joined the trends in the year 2004 by offering free email services that enable end-users to send messages virtually to their loved ones irrespective of their geographical locations in the world. Also, Amazon introduces Elastic Cloud computing (EC2) that enables numerous end-users to pay and use applications dynamically, on-demand, over the Internet [9]. In the year 2010, Cloud Azure was introduced by Microsoft for the delivery of services to end-users. These cloud services comprise storage space, applications and hardware. Over the last five years, great successes have been achieved towards integrating Cloud and IoT Technology as both tend to complement each other. IoT sensing devices generate enormous data and are prone to limited storage resources, processing power, and energy. Therefore, the cloud offers a practical solution to leverage IoT devices' challenges by providing its virtually unlimited resource(s) storage space and high



computational capabilities. Hence, "extending the scope of IoT to deal with real-world things for the delivery of new services in a large number of real-life scenarios" [10], which lead to the formulation of a new era in Information Technology, called Cloud-IoT. The new service paradigm includes Sensing as a Service (SaaS) for enabling virtual access to remote sensor data and Sensor Event as a Service (SEaaS) for distributing messaging services obtained from sensor event coverage. Furthermore, Ethernet as a Service (EaaS) for enabling virtual layer-2 connectivity to remote devices [11], and Video Surveillance as a Service (VSaaS) for providing ubiquitous access to recorded video and implementing complex analyses on the Cloud [12]. Fig. 1 depicts the architecture of WSN enabled Cloud-IoT infrastructure. According to Zhou et al. [13], Cloud-IoT infrastructure comprises four layers: Cloud Things service platform, Cloud Things Developer Suite, Cloud Things Operating Portal, and WSN. Sensory data requests obtained from end-users via the Cloud Things Operating Portal are deposited in the Cloud Things service platform, which is further dispatched to the WSNs to provide sensed data.

The sensed data obtained from the WSNs are retransmitted into the Cloud Things service Platform via

a wireless communication channel (e.g., WiFi and 4G) for further processing. The Cloud Things service platform is a set of cloud services (IaaS) that provides hardware (e.g., servers, storage, and network) and associated software such as operating systems virtualization and file system as services [14]. It interacts directly with the WSNs and enables the collection, storing and retrieving of sensory data. Furthermore, a massive sensory data volume is processed and analyzed using the computational resources on the Cloud. Cloud Things Developer Suite is a conventional cloud service tool (PaaS) that leverages the development platform to design, implement, and deploy its Applications [15]. An example of service tools consists of Application Programming Interfaces (APIs) based on the open-source web. Cloud Things Operating portal is a cloud service (SaaS) that supports the distribution and handling of specialized processing services that comprise the discovery of WSN physical features and data intelligence [13]. WSNs are tiny low powered chips deployed to monitor a specific geographical area or the internal organs of humans. It consists of four essential components: the processing unit, sensor, communication, and power units [16].



Figure 1. Architectural design of WSN enabled CloudIoT (Shows the different components that makeup the WSN-CLoudIoT Infrastructure)



## **3. RESEARCH METHODOLOGY**

The methodology framework is utilized to conduct this The literature covers research survey. search contributions from 2012 to 2019, exploring the most appropriate databases to find the targeted articles. Fig. 2 denotes the total number of previous works of literature obtained to actualize this study's objectives. These databases include Elsevier, Springer, IEEE Xplore Digital Library, Emeraldinsight, IGI Global and ACM. The search queries that were utilized to retrieve relevant literature are ("Algorithms for WSNs in Cloud Internet of Things) and ("Sensory Data management in Cloud internet of Things). Numerous literature relevant and irrelevant to the current study were retrieved based on the search queries-resulting in a total of 106 candidate articles obtained from the initial search. Furthermore, literature published only in English and contained in conference proceedings, white papers and journals were considered. Each candidate article passed through some critical stage of scrutiny before they were selected, which are listed as follows;

- Assessing the title of the candidate articles and removing those that are not relevant to the current study.
- Reading the abstract and discarding those unrelated to the study (algorithms used for the management of WSNs enabled Cloud-IoT).
- Read the introduction and conclusion to assess the quality of the contribution, and discard if the contribution quality is low.



Figure 2. Research Methodology (Describing the steps taken to obtain related literatures in this study)

The parameters used for the assessment of quality contribution are based on the level of relevance about the

existing literature with the algorithms deployed to manage WSNs-enabled Cloud-IoT infrastructure, soundness, credibility and writing quality. After which, a total of 30 articles are deemed to have passed the selection process as depicted in Fig. 2. Therefore, 30 articles were selected to actualize the research aim and the level of interest in utilizing algorithms for managing WSN's physical features and its sensed data in Cloud IoT infrastructure.

#### 4. **RELATED WORKS**

This section summarizes the research work of existing survey papers on Cloud-IoT to determine its present developmental potentials and aid the problem formulation of the research under study. A survey on sensor cloud infrastructure is designed by [17]. It discusses the general overview of the sensor cloud about its definition, architecture, and platform applications. It also presents the existing challenges, solutions, approaches and future research scope. The investigation of IoT Trust Management is formulated in [18]. It analyses relevant literature articles on the trustworthiness of WSNs. The issues yet to be resolved are highlighted and proposing architecture for the overall Trust Management on IoT system deployment. A review on techniques and state-of-the-art research work in IoT data-centric perspective is presented in [19]. It discusses various data storage models, complex event processing, comprehensive searching techniques and data stream processing in Cloud-IoT. The current IoT technology investigation by considering its physical layer, application, and data layer is formulated by [20]. It also focuses on fifth-generation (5G) networks and semantic web for IoT infrastructure. Furthermore, [21] introduced a research survey based on the management of various cloud services such as analysis tools, applications, servers and virtualization. Consequently, [22] embark on reviewing a high-level conceptual layered architecture of IoT- Cloud enabled Fog computing from a computational viewpoint. The architecture emphasizes issues related to cloud computing, fog, sensors/actuators and how they can be managed. The current research survey understudy aims to explore the existing algorithms deployed to manage WSNs physical features and its huge amount of sensed data in the Cloud-IoT platform, as opposed by previous research surveys in this field. The algorithms are classified by their processes namely Clustering, Classification. Oueuing and Meta-heuristic.

## 5. ANALYSIS OF THE EXISTING ALGORITHM

It is imperative to retrieve timely data from multiple WSNs to be transmitted onto the cloud data center on a real-time basis. The algorithms developed by previous researchers for managing the physical features of the WNSs and their heterogeneous sensed data are presented in this section.

## A. Clustering Process

The template is designed so that author affiliations are not repeated each time for multiple authors of the same affiliation. Please keep your affiliations as succinct as possible (for example, do not differentiate among departments of the same organization). This template was designed for two affiliations. It is an unsupervised learning process that extracts hidden patterns and structures from the dataset. Unlike classification with some prior knowledge to strategize the partitioning operation, clustering does not have any pre-knowledge about the strategy to be used for the extraction process. It aggregates the data into groups based on their similarity features and a standard structure and the data points in various dissimilar clusters [22]. However, the "fundamental challenge of clustering process or technique is based on its effectiveness in dealing with dynamic and heterogeneous sensing data partitioning to find optimal solutions for NP-hard problems" [23]. Therefore, it is mandatory to implement a suitable clustering algorithm that can effectively address the targeted problem, for instance, to solve combinatorial optimization sensory data in the Cloud-IoT infrastructure, where the computational processes become critical.

Implementing a clustered base algorithm to obtain sensing data from multiple WSNs is presented in [24]. The sensing data retrieved are transmitted to a local database server located close to the WSNs to minimize nodes' sensor network energy consumption. Consequently, the database server is decolonized and distributed to transparently and efficiently transmit the sensing data to the cloud data center. The clustering algorithm is responsible for abstracting sensing data and actuation resources from WSNs. It also facilitates the connection of virtualized sensor network nodes on the cloud with the Autonomic enforcer's support, enabling the physical features of the WSNs to be accessed virtually by end-users. A resource and bandwidth-aware wireless sensor (capsule) video summarization algorithm on mobile phone enabled cloud-based infrastructure is proposed in [25]. Wireless capsule sensors are digested in the form of a pill-sized capsule that allows the diagnostic image monitoring of the entire gastrointestinal (GI) digestive tract in humans. It is composed of a tiny camera, light source, radio frequency transmitter, and batteries. It pushes a large volume of video frames obtained into mobile phones where they are preprocessed. The summarization algorithm is responsible for extracting relevant video frames from multiple frames by eliminating irrelevant video frames. This is done by comparing the frames using Jeffrey-divergence between the histograms color and inter-frame correlation of the

adjacent video frames' color channels to discover the irrelevant video frames. The un-useful video frames are eliminated by extracting multi-fractal texture features using the ensemble-based clustering process. The actual video frames obtained are offloaded to the cloud data center for the onward process. Furthermore, Liu [26] proposed a Two-step K-means Clustering (TKMC) algorithm to cluster the image sensed data retrieved from IoT devices into two categories namely, Blurry and Clear Image. Thus, the blurry images are discarded while the clear Images await further processing. Clear image sensed data is further segmented into two categories, namely foreground (Containing the actual image data) and background (containing un-useful image data), by utilizing the watershed segmentation function at the edge (i.e., clusters of servers) layer. This is done by extracting the clear images and removes the background image leading to the actualization of the foreground (desired) image.

Productive Periodic-frequent Pattern-growth (PPFPgrowth) algorithm is proposed in [27] to reveal all the productive-associated regular, frequent patterns from sensed data. It further removes uncorrelated periodic data from the sensed data sets. In simple words, it can filter a large volume of periodic patterns to discover only the correlated ones, by utilizing large chunk of memory for storing only the transaction information (correlated) at the tail-end of each branch. Table I illustrates the clustering algorithms' comparison highlighting the challenges resolved and achievements obtained, together with their feature improvement.

Conversely, [28] proposed a Map-reduced Productive periodic-frequent Sensor Patterns (PPFSP-H) mining algorithm for efficient mining of the complete set of productive periodic-frequent sensor patterns among the discovered patterns from body sensor nodes. This is actualized by utilizing the pattern growth inspired function and a tree structure to identify and chronologically associate similar frequent patterns, reducing the cloud server's computational memory usage while improving the processing time. The development of a framework that queries massive audio/visual data obtained from an inbuilt mobile phone sensor camera device(s) is presented in [29]. Likewise, [30] implement a clustered-tree algorithm to reduce the energy consumption of visual sensor networks by considering their spatial-temporal coverage but unable to address the heterogeneous sensed data storage and the energy consumption challenges of the WSNs. However, the framework media query process comprises Pareto distributed/query volume, Exponential distribution, and Half-gaussian distribution algorithms. Thev are implemented to minimize the energy consumption of visual camera sensor networks and the billing cost of accessing visual sensory data from the



	Algorithms											
S/N	Article Title	Algorithm Name	Challenges	Achievement	Future Improvement							
1	Combining Cloud and sensors in a smart city environment [24]	i) Clustering Algorithm	The high rate of energy consumption in WSNs	Minimizes energy consumption	To deploy more efficient Filtering and Aggregation technique on sensory data							
2	Mobile-cloud Assisted Video Summarization framework for Efficient Management of Remote Sensing Data Generated by Wireless Capsule Sensors [25]	i)Summarization Algorithm ii)Ensemble clustering iii) Multi-fractal Texture Technique	Elimination of redundant video sensed frames, determining useful and un-useful video sensed data	Conserve the wireless capsule sensors energy consumption	To resolve the problem of Management of multi- objective optimization for the efficient provisioning of sensory data in Cloud- IoT.							
3	A new deep learning-based food recognition system for dietary assessment on an edge computing service infrastructure [26]	i) Two-step K-means Clustering (TKMC) algorithm	Over fitting issues due to massive data	Improved clustering accuracy with reduced response time	To develop a mechanism that will speed up the computation process.							
4	Mining productive- associated periodic- frequent patterns in body sensor data for smart home care. <i>Sensors</i> [27]	i) Lagrangian Productive Periodic- frequent Pattern- growth (PPFP-growth) algorithm	Usage of energy Filtering frequent data patterns from huge volume of sensed data	Shortest path route to Enhanced clustering process dud tot eh removal of recurrence data pattern from the data set.	To enhance it to a gene To develop a technique for resolving missing data issues.							
5	Mining of productive periodic- frequent patterns for IoT data analytics [28]	Productive periodic- frequent Sensor Pattern algorithm	The global search space of data pattern and the failure to explore the actual correlation between data.	Improved clustering accuracy while reducing memory usage and computation time	The implementation of a technique for improving accuracy while resolving missing data issues.							
6	Media Query Processing for the Internet-of-Things [29]	<ul> <li>i) Pareto Distributed- query Volume</li> <li>Algorithm</li> <li>ii) Exponential</li> <li>Distribution Algorithm</li> <li>iii) Half-Gaussian</li> <li>Distribution Algorithm</li> </ul>	trade-offs in the energy consumption and Billing cost Issue	Reduction of billing cost for provisioning of sensory data by providers, and minimizing energy usage	To enhance the expected performance under a broader range of parameter settings, using variations in the analytic modeling.							
7	Energy Consumption of Visual Sensor networks: Impact of Spatail-Temporal Coverage [30]	i) Clustered-tree Algorithm	Energy consumption problem of sensors with regards to their spatial-temporal coverage	Energy usage is conserved by limiting the sensing coverage	To adopt a mechanism that will efficiently manage the huge volume of sensed data in the cloud platform.							
8	Federated Internet of Things and Cloud Computing Pervasive Patient Health Monitoring System[31]	<ul> <li>i) Data</li> <li>Clustering/Classificati</li> <li>on Algorithms</li> <li>ii) Fuzzy Data Fusion</li> <li>algorithm</li> </ul>	Provisioning of real- time sensory data constraints and depletion of sensor networks lifetime	Minimize bandwidth transmission rate of sensor networks and improve battery lifetime	Privacy and Security Features are to be developed upon in the future as stated by authors.							

Table I. Shows the different types of Clustering and Classification algorithms for resolving the issues related to data redundancy generated from WSNs for the provisioning of meaningful and useful data in CloudIoT Infrastructure. It shows the results of the issues solved as well as improving the performance of the algorithms in future work



billing cost of accessing visual sensory data from the Cloud-IoT infrastructure. Additionally, analytic conditions were put forward to optimize the device(s) energy consumption and billing cost. This is done by determining the period at which sensing devices are in idle state. Sensed data are not generated during the idle state, which attracts minimum or no cost from cloud providers. Also, the time frame at which the sensing devices are in an active state (capturing audio/visual data) are monitored not to exceed the stipulated limits specified. Therefore, avoiding excessive energy usage by the sensors and the billing cost involved in storing and quarrying (processing) the sensed data generated. Implementing а pervasive Cloud-based WSNs infrastructure for remote monitoring of patient health in their respective homes is presented in [31]. It utilizes data clustering method to process sensing data retrieved from WSNs. Each sensor in the WSNs is capable of preprocessing by filtering the data (such as blood sugar, breathing, and glucose level as well as pulse oximetry and ECG rate) before dispatching them to a personal server via a blue tooth communication link. The clustering algorithm aggregates the sensed data, which triggers the alarm signal to notify the authorized patient and emergency healthcare personnel to access sensed data on the server on a real-time basis. The sensing data are also dispatched to the cloud for further processing and analysis. Moreover, the fuzzy-based data fusion algorithm is adapted to extract significant sensed data while discarding redundant ones at the local server located close to the WSNs. Hence, to conserve sensor nodes' energy usage and improve the bandwidth transmission rate between the WSNs and the cloud platform.

## B. Classification Process

Classification is a supervised machine learning process that assumes some prior knowledge to guide the partitioning operation to formulate a set of classifiers to represent the best distribution of patterns [32]. Furthermore, the classification technique is design to utilize labeled and unlabeled data during the classification process. The labeled data set is mainly used to train the classifier, such as the prediction function, while the classifier classifies the unlabeled data. The research work in [33] proposes a QuadTree Decomposition (QTD) algorithm for the partitioning of Multimedia sensed data (e.g., pictures) array. The sensed data are classified into two-dimensional matrixes of varying sizes (MBs), which are processed to avoid distortion reduction for each MB. It further prioritizes the MBs based on the outcome of their distortion reduction. The large flat area with multiple pixels is assigned with minimum large MBs, while the high-frequency edges with a minimum number of pixels are assigned with a large number of small MBs. Consequently, [8] develop a constraint programming-

based backpropagation algorithm to select the best route path to transmit sensed data from desired sensor nodes in the WSN. Hence, it reduces data packets' correlation between the selected desired nodes and inactive nodes, leading to enhanced accurate data classification. Furthermore, missing data are recovered from the inactive sensor nodes with the backpropagation technique's support. Also, [34] proposed a Discrete Wavelet Transform (DWT) algorithm to reduce excessive noise from sensed data signals to obtain relevant data signals. It removes the wavelet coefficients that contribute immensely to the noise from the signal. The relevant signal is retrieved by decomposing the sensor signals at various levels, after which vectors of coefficients are produced from each level. Therefore, the coefficient with the most wavelet is selected to reconstruct a new signal. A Collaborative-filtering-based Multi-criteria Matrix-based Integration (MCMLI) algorithm is developed in [35]. It determines the cohesive user-item subclasses for each criterion from sparse sensed data and forecasting the interests of users for each criterion accurately. Firstly, a user-item matrix is divided into a set of cohesive useritem subclass matrices, which are organized with correlated users and items for each criterion. This process is repeated until the subgroups accommodate all elements of the given user-item matrix. After which, user ratings on new items for each subclass are predicted. The predicted results are fused to make recommendations for users. Conversely, [36] introduced a Semantic Web Annotationbased Complex Event Processing (SWA-CEP) algorithm for the processing of IoT sensing data at gateway platforms. The SWA filters incoming sensor readings to discard irrelevant data with the support of its filter rule method. Then, the CEP processes the relevant sensed data by tagging them with shared vocabularies based on event rules. Hence, to facilitate the classification of the annotated relevant data into three types of events (normal, warning and critical level), they can be easily identified by higher-level applications and services.

Table II illustrates the comparison of classification algorithms deployed by previous researchers in this field. It comprise of the titles of previous researches based of classification processes and itemized the algorithms deployed to resolve the highlighted issues. Furthermore, it contains the benefits of the algorithms to with their weaknesses, which can be considered as future research directions in this field. Observation from the tables shows that the existing algorithms actually improve the performance of sensory data management by resolving the intending issues. Also, weakness of the research work in [33] was not specified in the literature. However, the other researches duly stated out the weakness of their proposed algorithms which are vividly itemized as denoted in the table. Observations also show that some of the algorithms were implemented at the cloud SaaS platform and the gateways. Consequently, the other ones were implemented in the cloud broker service layer.



	Algorithms											
S/N	Article Title	Algorithm Name	Challenges	Achievement	Future Improvement							
1	Multimedia Sensing as a Service (MSaaS): Exploring Resources Saving Potentials of at Cloud-Edge IoT and Fogs [33]	i) Quad Tree Decomposition (QTD) algorithm	High energy consumption during retrieval and sensed data from IoT sensor devices.	Conserve the energy usage of sensing Devices during sensed data retrieval sensed data.	NoT Specified							
2	Cloud-Assisted Data Fusion and Sensor Selection for the Internet of Things [8]	i) Constraint Programming-based Back-propagation (CP-BP) Algorithm	The issue of missing data that emanate from the high correlation of data between the inactive and active sensor	Enhances accuracy due to the proper selection of desired sensor nodes for retrieval of actual sensed data and the recovery of missing data.	To extend the proposed algorithm for real-time deployment. Also to integrate adaptive sampling rate into our current results, as well as investigating multi-sink scenarios.							
3	Insights Into IoT Data and an Innovative DWT-Based Technique to Denoise Sensor Signals [34]	i) Discrete Wavelet Transform (DWT) algorithm	The issue of excessive noise in sensed data signals.	Improves sensed data signal with 90 % accuracy of the reconstructed signal.	To enhance the proposed technique for forecasting purposes and to evaluate its accuracy by comparing the final decision-making process in realistic scenarios.							
4	Multi-criteria matrix localization and integration for personalized collaborative filtering in IoT environments [35]	i) Collaborative- filtering-based Multi-criteria Matrix localization and Integration (MCMLI) algorithm	Sensed Data-sparsity problem retrieved from users feedback	Improves recommendation accuracy.	Plan to develop a Bayesian non-parametric based matrix localization method to improve the performance of MCMLI.							
5	Event-driven and semantic-based approach for data processing on IoT gateway devices [36]	<ul> <li>i) Semantic Web Annotation (SWA) algorithm</li> <li>ii) Complex Event Processing Algorithm(CEP)</li> </ul>	High usage of computational resources, traffic bottleneck, and ineffective classification process.	Conserved computational resources and enhances transmission bandwidth rate from the gateways to the cloud. Also, enhances classification accuracy with response to time.	To investigate load balancing algorithms for managing the distribution of edge computing tasks based on resources available at the gateway level.							

TABLE II.	COMPARISON OF CLASSIFICATION PROCESS

Table II. This table highlights the problems solved by the existing algorithms that utilize classification process to manage the sensor devices and their data in WSN IoT-based Cloud infrastructure.

#### C. Queuing Process

Most queuing algorithms are based on evaluating the network feedback, based on network traffic congestion detection and taking precautionary steps to adjust the output source by minimizing the congestion window [37]. However, over-allocation of resources to be transmitted over a network channel, without considering the effect of buffers, can lead to traffic congestion and depletes the entire Cloud-IoT network performance. Fig. 3 depicts queuing characteristics algorithm. The development of a tailing dam monitoring and alarm systems-based Cloud-IoT is presented in [38]. Tailing is the remains of the milling process deployed to extract required metals from mined ores or cleaning of calm. The conventional monitoring system deployed for tailings monitoring is time-consuming and relies on humanitarian assistance to manually compute tailing operational parameters. However, the implemented Cloud IoT based tailing dam



Figure 3. Design of Queuing and Control algorithm (Illustrating how tasks are scheduled on a queue waiting execution)



monitoring and the pre-alarm system adopted the Regulatory based Control algorithm to resolve the conventional system's flaws. Wireless sensor nodes (drillhole pressure and water level, and the deformation sensors) are embedded at the tailing dam site to obtain data about the tailing dam's physical condition. Furthermore, the sensor nodes are connected via Wi-Fi to their respective sink node, which has an in-built microprocessor and transceiver. The sink node transmits data to a network layer that is saddled with the responsibility for receiving packets of data from multiple WSNs and dispatches its contents into the cloud to be processed into useful information. With the information in hand, the dry beach length and the top elevation levels are constantly monitored to prevent flooding of the tailing dam with the pre-alarm parameters' support. The Regulatory-based Control algorithm is also saddled with the responsibility to notify any unforeseen danger. For instance, if any sensor value is above the safety pre-alarm value, it triggers the alarm system for urgent response measures to be taken in order to prevent disaster. A multisensor gateway high-performance scheduling model of a cloud-enabled A sensor platform for a smart living is proposed by [39]. Scheduling and catching sensed data are executed by unifying 0-1 programming and periodic task algorithms before they are dispatched to the cloud. Earlier work in [40] introduces polynomial-time contestant-speed feasibility and weight maximization variant algorithms based on a communication-centric scheduling scheme. They are deployed to test for the multi-sensory platform and determine a subset of sensor request tasks with maximum weight that can be scheduled. The aforementioned communication-centric scheduling scheme only focuses on the media access control (MAC) layer, which is posed with the complexity to find the optimal channel for scheduling multiples of sensory request tasks. However, the multi-sensor gateway high-performance scheduling model is composed of a manager, primary manager (MM), a cache manager (CM) and the scheduling manager (SM) servers. The primary manager provided a unified input and output framework for retrieving sensed data from WSNs and dispatched them to the cache and scheduling managers. The cache manager stores sensed data temporarily and trigger the scheduler manager to schedule sensed data by executing the 0-1 programming algorithm based on end-user requests.

The development of a Cloud-IoT enabled architecture is presented in [41]. It is based on the utilization of HyperText Transfer Protocol (HTTP) and Massage Queuing Telemetry Transport (MQTT) protocols for optimal performance. The proposed architecture is composed of multiples of WSNs, router, and cloud data centers. The WSNs are embedded in the environment and transmit their sensed data via a router device to the cloud. The cloud data center consists of HTTP and MQTT servers. MQTT server is regarded as the master node, initiating other servers (e.g., HTTP server) as slaves. It uses its broker-based protocol to publish and subscribe to message transmission, enabling efficient interaction between the WSNs and end-users. It also maintains a steady use of Transmission Control Protocol (TCP) connection between the end-users and the WSNs, to improve real-time service performance and reduce the energy consumption of WSNs. Furthermore, it guarantees reliability by deploying three quality of service (QoS) acknowledgment/handshake, mechanisms: hypertext transfer protocol, and the Cloud-IoT platform. The acknowledgment and handshake mechanism are used to ensure that all requests are successfully delivered to prospective end-users. The hypertext transfer protocol utilizes a web-enabled application framework to interact with clients via a request-response cycle. Subscribed clients can easily access the HTTP server resources by sending a request via mobile phones. Upon receiving a request, it is acknowledged, processed, and the results retransmitted back to the clients. The Cloud-IoT platform uses a database cluster mechanism called redis2 to store data. Redis2 cluster with multiple Redis nodes is configured in the Cloud-IoT to enhance the database's reliability and the entire system. The implementation of an edge router deployed to interconnect multiple of WSNs is presented in [42]. Regressive Admission control (REAC) and Fuzzy Weighted Queuing (FWQ) algorithms are implemented in the edge router to monitor unforeseen changes in the network quality of multiple WSNs. Also, to provide efficient and stable network performance by optimizing network path and resources. The FQW algorithm improves the traffic path for the smooth transmission of sensed data from the WSNs to the edge router and cloud. On the other hand, the REAC algorithm manages the performance of end-to-end networks, supported by the real-time QoS monitoring algorithm for the prevention of malicious traffic or denial of service attacks on the networks, during data transmission phase. An intelligent queuing algorithm is also implemented in the edge router to prevent traffic congestion and efficiently allocate resources to specific traffic flows.

A Real-Time Adaptive Algorithm based on the Markov decision process (MDP) for Video Streaming over Multiple Wireless Access Networks based mobile Sensing Cloud platform is proposed by [43]. It utilizes parallel computation for real-time video streaming that enables mobile sensing devices to offload computation, communication, and resource-intensive operations to the cloud. On the other hand, the cloud is responsible for the computational processes and storing of Real-time Video Streaming (RTVS) frames. It is also saddled to forward processed RTVS to mobile sensor devices based on enduser requests, leading to the reduction of energy usage and optimal performance of the mobile devices. Also, more memory space is conserved to speed up other processes that run on mobile sensor devices. Conversely, [44] proposed a Secondary Exponential Smoothing (SESM)



Prediction algorithm for processing sensed data retrieved from WSNs. SESM monitors the sensed data traffic between WSNs and gateway to determine if there are numerous or limited sensory data obtained from sensor nodes. The sensor node is compromised when the observation result shows that sensed data traffic is below or above the expected acceptable benchmark at a particular time interval, which the network manager prevents. After this, the sensed data are filtered to obtain the actual data within the targeted threshold of the accepted benchmark range. The ones within the benchmark range are discarded while the actual sensed data are transmitted to the cloud for onward processing. The cloud gateway is saddled with the responsibility to compress end-users requests and decompress (end-user sensed data) sensed data received. Therefore, it minimizes packet loss due to network traffic and dispatches them into the cloud for further processing. The Efficient Power Saving Scheduling algorithm based on dynamic AID

allocation and Compressed Bitmap algorithms is presented in [45]. Thus, to reduce sensor metering devices' energy consumption by successfully scheduling the uplink and downlink traffic metering on the internet. Similar research work in [46] presented a power-saving with Offset Listening Interval algorithm to avoid collision by controlling the station wake-up time with the computed offset. However, the efficient power saving scheduling algorithm is enhanced to provide more lifetime to STAs battery of multiple WSNs, by utilizing the features such as small payload, low duty, and collision resolution technique. It also allows the STAs batteries to be allocated to the average service duration leading to STAs uninterrupted services. On the other hand, the Compressed Bitmap algorithm is adopted to reduce the communication overheads between the WSNs and cloud by compressing sensed data with similar features. Table III below shows the complementary information about the existing algorithms that utilize the queuing process.

	Algorithms												
S/N	Article Title	Algorithm Name	Challenges	Achievement	Future Improvement								
1	The internet of things (IOT) and cloud computing (CC) based tailings dam monitoring and pre-alarm system in mines [38]	i) Regulatory-based control algorithm	Monitoring of tailing dam site	Improves monitoring and pre-triggers alarm system to notify unforeseen danger	Unable to consider the tradeoffs between energy efficient of WSNs and Quality of data transmission, which has to be resolved in future work.								
2	A high- performance scheduling model for the multisensor gateway of cloud sensor system- based smart- living[39]	<ul><li>i) 0-1 programming algorithm</li><li>ii) Periodic algorithm</li><li>iii) Cache Technique</li></ul>	Optimal channel for scheduling multiple sensor requests and an NP-hard problem for optimizing scheduling on multiple-hop	Efficient channeling of numerous sensor requests and improves the allocation of a sensor node for task execution	To Consider the overhead from application preemption and prioritization of sensor nodes for the performance of sensory request tasks.								
3	Algorithms and complexity for periodic real- time scheduling [40]	i) Polynomial-time Constant-speed Feasibility Algorithm ii) Weight Maximization Variant Algorithm	Determination subset of sensor request tasks for scheduling	Multi-sensor request task was scheduled optimally based on the magnitude of their weight	To address the issue of a subsetof the sensory request tasks to be scheduled with minimal profit, by maximizing profit and optimal scheduling of multi-sensor tasks.								
4	Internet of things Cloud: Architecture and Implementation [41]	<ul> <li>i) Hyper Test Transfer</li> <li>technique (HTTP)</li> <li>ii) Message queuing</li> <li>telemetry transport</li> <li>(MQTT) technique</li> </ul>	Reliability of sensory data transmission and management of physical sensor nodes	Improves transmission of reliable sensed data and minimizes energy consumption of physical sensor nodes	Security and Privacy issues to be considered in future work.								
5	An Adaptive Edge Router Enabling Internet of Things [42]	<ul> <li>i) Regressive admission</li> <li>control (REAC) algorithm</li> <li>ii) Fuzzy weighted</li> <li>queuing (FWQ) algorithm</li> <li>iii) Intelligent scheduling</li> <li>queuing algorithm</li> </ul>	Monitoring network performance changes the prevention of malicious traffic	The transmission route path is Improved leading to enhanced bandwidth communication rate and efficient allocation of sensory resources.	Fusion of numerous algorithms for enhancing the performance of WSNs resources used for the servicing of applications.								

TABLE III. COMPARISON OF QUEUING PROCESS

			Algorithms				
S/N	Article Title	Algorithm Name	Challenges	Achievement	Future Improvement		
6	Design and implementation of a novel service management framework for IoT devices in the cloud [43]	i) A real-time adaptive algorithm based on Markov Decision Process (MDP)	processing of video stream from multiple WSNs	Improves processing and efficient delivery of processed sensed video stream to consumers	Allocation of sensing resources based on scheduling approach for segmentation of video stream data in various surrogate modes and the forecasting of available bandwidth for streaming		
7	A novel sensory Data Processing Framework to Integrate Sensor Networks with Mobile Cloud [44]	i) Secondary exponential smoothing (SESM) prediction algorithm	Monitoring and prediction of data traffic	minimizes data traffic congestion resulting in minimal data packet loss during transmission	Not Specified		
8	An Efficient power saving polling scheme on the internet of energy [45]	<ul><li>i) Dynamic AID allocation algorithm</li><li>ii) Compressed bitmap algorithm</li></ul>	Uplink and downlink traffic metering control	Reduced communication overhead and prolong the lifetime of WSNs Batteries	To manage WSNs data resources under ultra-dense networks, considering the congestion channelingto handle multiple access points procedure.		
9	Power Save with Offset Listen Interval for IEEE 802.11ah Smart Grid Communications [46]	i) Offset Listening Interval Algorithm	Collision offset problem between sensor nodes	Avoid collision by controlling the station wake-up time with computed offset prolong the lifetime of sensor nodes	Not Specified		

Table III. This table shows the algorithms that adopted the queuing process for the scheduling of sensory request tasks and sensor nodes for efficient dissemination of sensory data in the Cloud-IoT infrastructure. It also shows the results of the problems resolved and future improvement on the performance of the algorithms.

## D. Meta-heuristic/Hybrid Process

Meta-heuristic is a strategy that makes some modifications in the heuristics to give better solutions in local search optimally [47]. They deliver suitable substitution between the highly-rated solution and computing time for resolving complex problems. Solutions obtained from heuristic methods are usually caught up in a local minima problem, which is resolved by the deployment of meta-heuristic algorithms. Consequently, meta-heuristics are often implemented to determine the best solutions from among multiple possible ones. However, they do not ensure that the best will be actualized in some instances. In most cases, they efficiently determine a solution close to the desired one. The hybrid process combines two or more algorithms to resolve a specific problem [48]. This is done either by selecting one of the more suitable algorithms or swapping them throughout the execution process. A task execution framework called Task-Alloc for Cloud-enabled WSNs is developed in [49], to reduce energy consumption usage of newly added user applications. It enables the selection of desired sensor nodes in the WSNs for sensed data retrieval and allows the sensed data obtained from a single node to be shared by multiple application requests. Previous research work in this field is challenged with an increase in WSNs workload. The energy of sink nodes drains out quickly within a limited duration due to an increase in the number of sensing request tasks on the

network. However, the Task-Alloc framework introduces two algorithms namely Hot-Tasking and Merge-OPT, to resolve the challenges mentioned above. The Hot-Tasking algorithm enables selecting desired sensor nodes for the retrieval of sensed data while other sensor nodes are set in sleep mode to conserve their energy usage. Suppose there are multiple similar requests from a single sensor node. In that case, the Merge-OPT (Based on Greedy Algorithm) algorithm is executed to merge them to become a single request for optimal sensed data retrieval from the desired node. Performance evaluation shows that both algorithms have the potential to conserve more than 76% energy usage of the WSNs when a new application request is added. An efficient Particle Swarm Optimization combined with Simulated Annealing (PSOSAA) algorithm is proposed in [50] for medical monitoring and managing cloud-based IoT infrastructure applications. The proposed algorithm is implemented in a middleware residing on the cloud resource database. The particle swarm optimization algorithm searches for the actual sensor node among multiple seamless sensor nodes that can deliver the required data as requested by end-users, considering the proximity between potential sensor nodes. On the other hand, the simulated annealing algorithm is implemented to support the particle swarm optimization by increasing its searching ability in order to obtain the best sensor node required for the acquisition of sensed data in real-time for the continuous monitoring of patient



health. In [51], a collaborative scheduling algorithm for a mobile sensor-enabled cloud system is implemented to minimize the number of activated mobile sensing devices. Sensed data from a specific mobile device is used to satisfy the request of multiple applications. The research work is based on a middleware that mediates between multiples of applications requesting sensed data and mobile sensing devices. The energy consumption of mobile sensor devices is minimized with the No-Aggregation algorithm implemented in the middleware. Also, the Info-Aggregation algorithm determines the tradeoff between energy efficiency and the amount of sensed data to be offloaded by active mobile sensing devices onto the IaaS cloud. The middleware regularly decides which mobile sensing device to activate from the multiple networked mobile sensor devices on a real-time basis. Hence, the desired sensed data are transmitted to the Cloud for further processing and storage. Table IV denotes the various types of algorithms that utilize the meta-heuristic and hybrid process in this research.

A closed-loop design model based on the Genetic algorithm for Cloud assisted IoT middleware is proposed in [52], to monitor the condition of multi-manufacturing engineering systems. System design weaknesses, failure, and predicting system health status are monitored on a real-time basis to proffer a timely solution to improve the system redesign process and performance status. WSN nodes are embedded in the manufacturing system components situated in different locations for operational data gathering (precision, temperature, and production speed) to determine their condition in real-time. The operational sensed data are transmitted via Bluetooth communication network from the multiple WSNs to gateways. Consequently, the sensed data are preprocessed in the gateways before they are offloaded to the cloud. The cloud is responsible for storing all the sensed data received from multiple WSNs, evaluate and translates them into knowledge base information (e.g., components weaknesses and failure, production speed and precision). Hence, to effect timely improvement, minimize

TABLE IV.	COMPARISON OF META-HEURISTIC/HYBRID PROCESS
-----------	---

	Algorithms												
S/N	Article Title	Algorithm Name	~		Future Improvement								
1	A Task Execution Framework for Cloud-Assisted Sensor networks [49]	i) Hot-Tasking Algorithm ii) Merge-OPT based greedy algorithm iii) Merge-OPT based Hybrid algorithm	Challenges Increase workload of sensor nodes	Achievement Merging of same multiple sensing requests satisfied by a single sensor node resulting in energy conservation	To investigate and formulate a security mechanism to prevent potential threats in the future.								
2	The monitoring and managing application of cloud computing based on Internet of Things [50]	<ul><li>i) Efficient particle swarm optimization algorithm</li><li>ii) Simulated Annealing algorithm</li></ul>	Searching for a suitable sensor among multiple seamless sensors considering their proximity	Improves searching capability thus obtained actual sensor required to fulfill the on-demand desired request.	To resolve the strict mathematical proof of convergence issues with an efficient algorithm, while giving priority to sensory task to be scheduled before they are dispatched for sensing event.								
3	Energy-aware collaborative sensing for multiple applications in mobile cloud computing", Sustainable Computing [51]	<ul><li>i) No-aggregation algorithm</li><li>ii) Info-aggregation algorithm</li></ul>	The Trade-off between energy efficiency and sensed data offloading	multiple requests satisfied by a single mobile sensing device while others remained in-active resulting in energy conservation	Future work is to address the provisioning of applications with various options for partial covering and reliable covering of requested regions by mobile devices.								
4	Closed-loop design evolution of engineering system using condition monitoring through the internet of things and cloud computing [52]	i) Genetic algorithm	Monitoring and provisioning of sensing data	Improves monitoring and efficient provisioning of timely sensed data	To Investigate the systematic approach in WSN-Cloud for the extraction of related features of sensing for detecting potential manufacturing design weaknesses.								

	Algorithms												
S/N	Article Title	Algorithm Name	Challenges	Achievement	Future Improvement								
5	A collaborative resource management for big IoT data processing in Cloud [53]	<ul><li>i) Non-dominating sorting-2</li><li>ii) Pareto strength evolutionary-2 model</li></ul>	Selection of suitable sensor resource providers with efficient sensor networks energy consumption	Optimized selection of desired sensor resource providers for execution of task requests and prolonging the lifetime of sensor network nodes	To deploy Bee and Ant colony algorithms or a combination of both to enhance the partner selection problem in the future.								
6	DEED: Dynamic Energy-Efficient Data offloading for IoT applications under unstable channel conditions [54]	Dynamic Energy Efficient Data offloading Scheduling algorithm	Unstable transmission channel conditions of offloading sensed data from IoT sensing device(s) to the fog and cloud for storage.	Enhances the reliability of transmitting sensed data via the desired communication channels to the cloud while conserving energy usage.	To study the issue of fault tolerance model for the offloading of sensed data in order to further improve the data reliability and investigate the channel quality-awareness with respect to Zigbee protocol for the development of distributed scheduling techniques.								
7	Workload-aware VM consolidation method in edge/cloud computing for IoT applications [55]	Workload-Aware Virtual Machine Consolidation (WAVMCM) algorithm	The inability for edge devices to process tasks in a power saving mode.	Minimizing the number of active server utilization leading to energy conservation.	To determine how to minimize the use of hazardous materials, heat generation and cloud computing resource wastage.								
8	Multimedia Sensing as a Service (MSaaS): Exploring Resources Saving Potentials of at Cloud-Edge IoT and Fogs [32]	Truncation-Point- Optimized Resource Allocation (TPORA) algorithm	The issue of bandwidth scarcity during sensed data offloading to the cloud	Effective allocation of optimal bandwidth rate for the offloading of sensed data into the cloud storage server.	Not Specified								

Table IV. This table illustrates the comparison of previous algorithms that adopted meta-heuristic and hybrid process for resolving issues of the WSNs resources in Cloud-IoT, thus showing their results and future improvement about the performance of the algorithm.

production cost and increase productivity. In [53], a global collaborative cloud data center for resource management provision is proposed to process numerous and heterogeneous IoT's data set. The research aims to determine suitable means for targeting cloud providers or data centers across the globe that will form a cloud confederation. It utilizes multi- genetic algorithms namely Non-dominated Sorting-2 and Pareto Strength, to optimize selection among various cloud providers with the desired WSNs. The genetic algorithms optimize the selection by utilizing the profit gain and trust relationship information of individual cloud providers that intend to join the cloud confederation. Also, a trust model for the cloud provider selection that uses both objective and subjective trust values is adopted to confirm the trustworthiness of the collaborating cloud providers, ensuring the validation of cloud providers involved in the Truncation-Point-Optimized collaboration. Resource Allocation (TPORA) algorithm is developed in [32], to allocate communication resource (bandwidth rate) to the sensed data, based on their level of quality, without considering the unstable channelization conditions. Optimal bandwidth rate is allocated to premium packets sensed data compared to regular sensed data to be

offloaded in the cloud. However, [54] develop a Dynamic Energy Efficient Data offloading Scheduling algorithm. It resolves unstable channelization communication during offloading sensed data from IoT constrained devices to the cloud. Duplicate sensed data records retrieved from the sensor nodes are backed up to ensure reliability. After that, the earliest and expected time are computed to determine the possible time a device (e.g., mobile phones) in the network can offload its sensed data to the cloud. The device with the least computed time among other devices can offload its data on the cloud before others. Therefore, the constrained devices' energy usage is drastically reduced by selecting the device whose energy exceeds the threshold value to transmit sensed data. In contrast, others are put on sleep mode. Consequently, [55] implemented a workload-aware Virtual Machine (VM) consolidation algorithm to switch the idle physical server into hibernation mode. The resources of the physical machine are classified into four classes with diverse resource capacities to execute different VM requests. At the initial stage, new VM request tasks are aggregated depending on their resource demand. After which, they are allocated to a VM class that is capable of storing the sensed data. Therefore, VMs from low load

365



servers are migrated to intermediate load servers within the same class and determine the physical servers inactive mode, thereby hibernating them to minimize power consumption.

#### 6. EFFICIENT QOS DELIVERY PARAMETERS

The WSNs constrained issues, such as the inefficient transmission of sensory data resource (s) to the IaaSenabled cloud data center that leads to the energy consumption of WSNs, has been resolved significantly by the existing algorithms. The algorithms' effectiveness and efficiency were evaluated and validated based on metric parameters of Quality of Service (QoS) delivery in the Cloud-IoT platform. These parameters include Sensor Quality, Energy, Accuracy, and Throughput, Cost efficiency, Priority, Latency, Bandwidth rate and Data loss recovery [56]. Sensor quality is measured by the level of its sensitivity and accuracy of covering a specific region within a given time [57]. Energy Consumption is computed with regards to a particular energy model. For instance, the determination of the total volume of data transmitted. Therefore, energy usage directly influences the lifetime of any sensor network. Throughput is sensor nodes' ability to transmit their sensory data over a link to their connected sink nodes within a given period [58]. Accuracy is the level of performance obtained during sensed data mining or analytics (management). Cost efficiency determines the sensor-cloud capacity to efficiently allocate limited resources to execute sensing request tasks that leads to optimized resource utilization. Data recovery refers to accessing logically or physically damage data without any functioning backup [59]. Priority is the means of given precedence to a specific task for execution before others. In WSNs, precedence is given to suitable sensory nodes that can deliver the desired sensing data requests. Latency determines the timeframe delay in the sensory network during the sensing task process. The bandwidth of physical sensor nodes is the measurement of bit-rate available for communication between sensor nodes to sink nodes and to the cloud IaaS platform. Table V depict the authors and their respective algorithmic processes used to actualize the efficient quality of service delivery (QoS) parameters. In table V, column three illustrate the Cloud-IoT hardware components which comprise of cloud data storage/virtual machines, and physical sensors, deployed by researchers for the actualization of resource management in Cloud-IoT infrastructure. Furthermore, different gateways are also used to manage data transmission by reducing the communication distance between the WSNs and cloud data center while improving the communication speed. The measurement of sensor quality is classified into two types, namely lightweight and heavyweight. The lightweight sensors are mostly deployed for capturing data signals, while the heavyweight is used for capturing multimedia data signals such as images and live video streaming from a camera fitted sensor nodes to the cloud The level of efficient QoS delivery IaaS platform.

actualized by the existing algorithms as presented in Fig. 4. The chart shows that sensor node quality has the greatest consideration than other parameters. It also shows that clustering and meta-heuristic algorithms have the highest implementation of QoS parameters based on the level of each parameter. However, throughput is mostly considered for the validation of meta-heuristic algorithms. On the other hand, energy happens to have the highest level of usage by the queuing algorithms. Furthermore, throughput and data recovery are least considered for the performance validation of clustering, classification and queuing algorithms. Accuracy is highly used to validate the performances of clustering and classification algorithm without its consideration in meta-heuristic algorithms. The cost efficiency determines the minimum amount of resources (such as the bandwidth rate, number of sink nodes and other gateways) use in transmitting sensing requests from the cloud data center to various WSNs. Also, the retransmission of sensed data requests from the WSNs back to the cloud. Furthermore, the computation tasks such as data gathering and the aggregation of multiple sensed data are efficiently performed in the cloud platform. The minimum time required for sensing coverage was also actualized, which is termed as latency for selecting the desired sensor nodes for sensed data delivery. In a nutshell, QoS contributed immensely for the actualization of efficient sensed data delivery.





S/N	Comparison of Parameters														
					Gateway	s				Par	ameters	s of QoS			
	Algorithms	Reference	Cloud-IoT Componen t (VMs, WSNs, Servers)	Smart Device	Sink Node	Cloud Broker	Sensor Quality	Band width Rate	Energy	Through put	Cost Eff.	Loss Data Recovery	Priory	Latency	Accuracy
		Mitten et al.[24]	Yes		Yes	Yes	Light weight	Less			Yes				
		Mehmood et al. [25]	_	Yes			Heavy weight				_	Yes	Yes		
	Clustering Process	Liu et al. [26]	_			Yes	Heavy weight				Yes		Yes		Yes
Α		Ismail and Hassan [27]	_		Yes		Light weight		Yes				Yes		Yes
		Ismail et al. [28]	_			Yes	Heavy weight								Yes
		Renna et al.[29]				Yes	Light weight		Yes				_		
		Redondi et al. [30]	_	Yes			Heavy weight		_	Yes				Yes	Yes
		Abawajy et al.[31]	_	Yes		Yes	Light weight				Yes		Yes		
		Wang et al. [33]	_	Yes			Heavy weight	_			_				Yes
в		Madhavaiah et al. [8]	_		Yes		Light weight		Yes				Yes		Yes
	Classification Process	Bijabooneh et al.[34]	Yes		Yes	Yes	Light weight	Less							
		Faria et al. [35]	_			Yes	Heavy weight								Yes
		Ko et al. [36]	_		Yes		Light weight	_	Yes				Yes		Yes

## TABLE V. EFFICIENT QUALITY OF SERVICE (QoS) DELIVERY PARAMETERS IN WSN-CLOUD-IOT

*E-mail: eaedje@graduate.utm.my, shafie@utm.my* 



S/N		Comparison of Parameters													
					Gateway	<b>'S</b>				Р	aramete	ers of QoS			
	Algorithms	Reference	Cloud-IoT Component (VMs,WSNs, Servers)	Smart Device	Sink Node	Cloud Broker	Sensor Quality	Band width Rate	Energy	Thro. put	Cost Eff.	Loss Data Recovery	Priory	Latency	Accuracy
		Sun et al.[38]	Yes			Yes	Heavy weight	Less			Yes				
		Lyu et al. [39]	_			Yes	Heavy weight	_	Yes	Yes			Yes		
		Bonifaci [40]				Yes	Light weight	_							
G	Queuing Process	Hou et al.[41]				Yes	Light weight	_	Yes						
C		Jutila [42]	_		Yes		Light weight	_							Yes
		Kumar et al.[43]	_			Yes	Light weight	_			Yes				
		Zhu et al. [44]	_			Yes	Light weight	_	Yes				Yes		
		et al. [45]	Yes		Yes		Light weight	_	Yes				Yes		
		Liu et al. [46]				Yes	Heavy weight	_	_	Yes				Yes	
		[49]		Yes			Light weight	_	Yes	Yes			Yes		
D	Meta-	Lou and Ren [50]	Yes		Yes	Yes	Light weight	Yes		Yes			Yes	Less	
	heuristic Process	Loomba et al.[51]	_	Yes			Light weight	_	Yes					Less	
		Xia et al. [52]	_		Yes		Heavy weight	Yes		Yes	Yes		Yes		
		Alelaiwi [53]	_			Yes	Light weight				Yes				
		Yan et al. [54]	_	Yes			Heavy weight	_	_	Yes				Less	

S/N							Compariso	n of Para	meters						
					Gateways			Parameters of QoS							
	Algorithms	Reference	Cloud-IoT Component (VMs,WSNs, Servers)	Smart Device	Sink Node	Cloud Broker	Sensor Quality	Band width Rate	Energy	Through put	Cost Eff.	Loss Data Recovery	Priory	Latency	Accuracy
N	Meta-	Mohiuddi and Almogren [55]	_	Yes		Yes	Heavy weight	Less	Yes		Yes	Yes	Yes		
	heuristic Process	Wang et al.[32]	_				Light weight	_		Yes	Yes		Yes		

Table V. It shows the parameters used to evaluate and validate the algorithms in order to determine their feasibility and performance. Also, it indicates that the existing algorithms mainly resolve the energy issue of the WSNs by using the computation power and storage features of the cloud to execute and store the huge amount of sensed data retrieved from the WSNs.



#### 7. DISCUSSION AND CONCLUSION

It is estimated that the Internet will be composed of multi-billion connected devices, leveraging their storage and computation processes with cloud platforms' support across the globe. These connected devices include Radio Frequency Identification (RFID), wired and Wireless Sensor Network (WSN) nodes, actuators and mobile sensing devices. Network communication channels from IoT devices to sink nodes and gateways and the cloud data center are gaining developmental stride daily. For instance, the LTE and 5G network communication technology were recently introduced to improve the communication overheads between IoTs connected devices and the Cloud IaaS platform. Though, the existing algorithms were able to resolve some of the challenges of managing WNS physical features and the data they generate to reasonable extents. However, they are not designed to handle the dynamicity and heterogeneity of big sensed data on a real-time basis, which is highly required in IoT. WSNs are known to generate a massive volume of data that conventional machine learning algorithms cannot manage. However, the combination of clustering techniques has been proven to be useful for handling big data in previous research, but it still requires further improvements regarding time computation complexity issues. For instance, delay in execution time and high cost of computation that leads to high usage of memory and storage resource(s). Resolving this problem will enhance clustering techniques to obtain better extraction of meaningful and desired data from the sample dataset. In the aspect of WSN, the existing algorithms mainly tend to reduce their energy usage with or without emphasis on how they can be deployed in a hostile environment for optimal and reliable sensing or event This can be handled with more advanced capturing. mining algorithms such as recommender systems and neural networks, together with sensor devices capable of capturing, pre-processing, and transmitting data or events to a local server or directly to the cloud for onward processing. As opposed to WSN nodes that are constrained with low memory and storage resource(s) and not transmitting the data captured directly to sever or the cloud platform.

In conclusion, an overview of the existing algorithms was analyzed on how they managed WSNs physical features virtually and its sensed data generation in the Cloud-IoT infrastructure. Consequently, the algorithms' future enhancement for better performance, as stated by previous research, is highlighted, which paves the way for future research directions. Detailed background information is presented about the origin and present state-of-the-art IoT and Cloud, as well as the factor that resulted in their integration. Also presented in this paper is the actualization of notable QoS parameters to validate the performance of the algorithms for the delivery of sensory data, the conservation of energy usage in WSNs and optimal usage of memory/computation processes in the Cloud IaaS platform. However, there are other potential issues to be addressed for the optimal performance of the Cloud-IoT platform, considering security and privacy. In the future, a specific meta-heuristic algorithm that has not been used previously will be designed and implemented to manage IoT devices.

#### ACKNOWLEDGMENT

Special thanks to Dr. Alade Abel, for his support and contributions. The Authors will also use this medium to thank the Research Management Centre (RMC) Universiti Teknologi Malaysia for supporting this study, with research grant (QJ130000.2451.07G48).

#### REFERENCES

- Kailas Devadkar and PrarthanaSawant. "Internet of Things and Integration with Cloud Computing", International Journal of New Innovations in Engineering and Technology. India, vol.6, pp.1-8, November 2016.
- [2] Mineraud, OleksiyMazhelis, Xiang Su and SasuTarkoma. "A gap analysis of Internet-of-Things platforms", Computer Communications (Elsevier). Netherlands, vol. 8-90, pp.5-16, September 2016.
- [3] Bhaskar Prasad Rimal, AdmelaJukan, DimitriosKatsaros, and Yves Goeleven. "Architectural Requirements for Cloud computing Systems: An Enterprise Cloud Approach", Journal of Grid Computing (Springer Link). Netherlands, vol. 9, pp. 3-26, January 2011.
- [4] Dave Evans. "The Internet of Things How the Next Evolution of the Internet Is Change Everything. CISCO White Paper", <u>https://www.cisco.com/c/dam/en\_us/about/ac79/docs/innov/IoT\_I BSG\_0411FINAL.pdf> [2018-10--12]</u>, America, pp.1-17, August 2011.
- [5] ShahabTayeb, ShahramLatifi, andYoohwan Kim (2017). "A survey on IoT communication and computation frameworks: An industrial perspective." IEEE 7<sup>th</sup> Annual Computing and Communication Workshop and Conference, America, pp.1-6, January 2017.
- [6] HaraldSundmaeker, Patrick Guillemin, Peter Friess and Sylvie Woelffle (2010). "Vision and Challenges for Realising the Internet of Things", CERP-IoT- Cluster of European Research Projects on the Internet of Things. Germany, pp.1-230, March 2010.
- [7] WaltenegusDargie and Christian Peollabauer (2010).
   Fundamentals of Wireless Sensor Networks: Theory and Practice, 2nd. ed., vol. 6. Wiley Publishing: 2010, pp.6-33.
- [8] Madhavaiah C., Irfan Bashir and Syed IrfanShafi. "Defining Cloud Computing in Business Perspective: A Review of Research", The Journal of Business Perspective (Sage). United Kingdom, vol.16, pp.163-173, July 2012.



- [9] Frank Cervone H. "An Overview of Virtual and Cloud Computing", International Library Perspectives (EmeraldInsight). United Kingdom, vol.26, pp.162-165, August 2010.
- [10] AlessaBotta, Walter de Donato, Valerio Persico, and Antonio Pescape. "On the Integration of Cloud Internet of Things", IEEE International Conference on the Future Internet of Things and Cloud. Bacelona, pp.24-30, June 2014.
- [11] ArkadyZaslavsky, Charith Perera and Dimitrios Georgakopoulos (2013). "Sensing as a Service and Big Data", Proceedings of the International Conference on Advances in Cloud Computing (ACC). Bangalore, pp.1-8, April 2013.
- [12] Andrea Prati, Roberto Vezzani, Michele Fornaciari and Rita Cucchiara. "Intelligent Video Surveillance as a Service", Intelligence Multimedia Surveillance (Springer Link). Germany, pp.1-16, July 2013.
- [13] Zhou J, Leppanen T, Harjula E, Ylianttila M, Ojala T andYu .C. "Cloudthings: A common architecture for integrating the internet of things with cloud computing. Computer Supported Cooperative Work in Design (CSCWD)", IEEE 17<sup>th</sup> International Conference. Canada, pp.651-657, June 2013.
- [14] Bhardwaj S, Jain L, Jain .S."Cloud computing: A study of infrastructure as a service (IAAS)", International Journal of Engineering and Information Technology. India, vol. 2, pp.60-68, April 2012.
- [15] Chinyao, Low, Yahaueh Chen, and Mingchang Wu. "Understanding the determinants of Cloud Computing adoption", Industrial Management & Data System (Emeraldinsight). United Kingdom, vol. 111, pp.1006-1023 August 2011.
- [16] Akyildiz, W. Su, Y. Sankarasubramaniam and E. Cayirci. "Wireless Sensor Networks: A survey", Computer Networks (Elsevier). Netherlands, vol. 38, pp.393-422 February 2012.
- [17] Daniele Miorandi, Sabrina Sicari, Francesco De Pellegrini, and Imrich Chlamtac. "Internet of Things: Vision, applications and research challenges", Ad Hoc Networks (Elsevier). Netherlands, vol. 10, pp.1497-1516, September 2012.
- [18] Everton Cavalcante, Jorge Pereira, Marcelo PitangaAlves, Pedro Maia, RoniceliMoura, Thais Batista, Flavia C. Delicato and Paulo F. Pires. "On the interplay of the Internet of Things and Cloud Computing: A systematic Mapping study", Computer Communications (Elsevier). Netherlands, vol. 89-90, pp.17-33, September 2016.
- [19] Yongrui Qin, Quan Z. Sheng, Nickolas J. G. Falkner, Schahram Dustder, Hua Wang and Athanasios V. Vasilakos (2016). "When things matter: A survey on the data-centric internet of things", Journal of Network and Computer Applications. America, vol. 64, pp.137-153, November 2016.
- [20] Anne H. Ngu, Mario Gutierrez, Vangelis Metsis, Surya Nepal and Quan Z. Sheng. "IoT Middleware: A Survey on Issues and Enabling Technologies", IEEE Internet of Things. Canada, vol. 4, pp.1-20, February 2017.
- [21] ParthaPratim Ray. "A Survey of IoT Cloud Platforms", Future Computing and Informatics Journal (Elsevier), Netherlands, vol. 1, pp.35-46, March 2017.
- [22] Amir Ahmad and LipikaDey. "A k-means clustering algorithm for mixed numeric and categorical data", Data & Knowledge Engineering (Elsevier). Germany, vol. 63, pp.503-527, November 2007.
- [23] Anil K. Jain. "Data clustering: 50 years beyond-means", Pattern Recognition Letters (Elsevier), Netherlands, vol. 31, pp.651-666, June 2010.
- [24] Nathalie Mitton, SymeonPapavassiliou, Antonio Puliafito, andKishor S. Trivedi. "Combining Cloud and sensors in a smart city environment", EURASIP Journal on Wireless Communications and Networking (Springer), America, vol. 4, pp.1-10, January 2012.

- [25] IrfanMehmood, Muhammad Sajjad, and Sung WookBaik. "Mobile-cloud Assisted Video Summarization framework for Efficient Management of Remote Sensing Data Generated by Wireless Capsule Sensors", Sensors (MDPI Journals). America, vol.14, pp.7113-7145, September 2014.
- [26] Liu, C., Cao, Y., Luo, Y., Chen, G., Vokkarane, V., Yunsheng, M. and Hou, P. "A new deep learning-based food recognition system for dietary assessment on an edge computing service infrastructure". IEEE transactions on services computing. Italy, vol. 11, pp. 249-261, October 2018.
- [27] Ismail, W., & Hassan, M. "Mining productive-associated periodic-frequent patterns in body sensor data for smart home care", MDPI Sensor. America, vol. 17, pp. 1-19, April 2017.
- [28] Ismail, W. N., Hassan, M. M., & Alsalamah, H. A. "Mining of productive periodic-frequent patterns for IoT data analytics", Future Generation Computer Systems (Elsevier). Netherlands, vol. 88, pp.512-523. November 2018.
- [29] Francesco Renna, Joseph Doyle, VssileiosGiostsas and YiannisAndreopoulos. "Media Query Processing for the Internetof-Things: Coupling of Device Energy Consumption and Cloud Infrastructure Billing". IEEE Transactions on Multimedia, vol. 18, pp. 2537-2552, July 2016.
- [30] Alessandro Redondi, DujdowBuranapanichkit, Matteo Cesana, Marco Tagliasacchi and YiannisAndreopoulos. "Energy Consumption of Visual Sensor networks: Impact of Spatio-Temporal Coverage", IEEE Transactions on Circuits and Systems for Video Technology. America, vol. 24, pp.2117-2131, May, 2014.
- [31] Abawajy, J. H., & Hassan, M. M. "Federated internet of things and cloud computing pervasive patient health monitoring system", IEEE Communications Magazine. Germany, vol. 55, pp.48-53, September 2017.
- [32] Wang, W., Wang, Q., and Sohraby, K. "Multimedia sensing as a service (MSaaS): Exploring resource saving potentials of at cloudedge IoT and fogs", IEEE Internet of Things Journal. Italy, vol. 4, pp. 487-495, August 2017.
- [33] Farshid Hassani Bijarbooneh, Wei Du, Edith C.-H. Ngai, Xiaoming Fu and Jiangchuan Liu. "Cloud-Assisted Data Fusion and Sensor Selection for the Internet of Things", IEEE Internet of Things Journal, Sweden, vol. 3, pp.257-268. April 2016.
- [34] De Faria, M. L. L., Cugnasca, C. E., and Amazonas, J. R. A. Insights into IoT data and an innovative DWT-based technique to denoise sensor signals. IEEE Sensors Journal. Switzerland, vol. 18, pp.237-247, January 2018.
- [35] Ko, H.-G., Ko, I.-Y., & Lee, D. "Multi-criteria matrix localization and integration for personalized collaborative filtering in IoT environments", Multimedia Tools and Applications. Canada, vol. 77, pp.4697-4730, June 2018.
- [36] Al-Osta, M., Bali, A., &Gherbi, A. "Event-driven and semanticbased approach for data processing on IoT gateway devices", Journal of Ambient Intelligence and Humanized Computing. United Kingdom, vol. 2, pp.1-16, April 2018.
- [37] Ali Ahammed and ReshmaBanu. "Analyzing the Performance of Active Queue Management Algorithms", International Journal of Computer Networks & Communications (IJCNC). America, vol. 2, pp.1-19, January 2010.
- [38] Enji Sun, Xingkai Zhang and Zhongxue Li. "The internet of things (IOT) and cloud computing (CC) based tailings dam monitoring and pre-alarm system in mines", Safety Science (Elsevier). Netherlands, vol. 50, pp.811-815, August 2012.
- [39] YongqiangLyu, Fanxi Yan, Yu Chen, Dan Wang, Yuanchun Shi, and Nazim Agoulmine. "A high-performance scheduling model for the multisensor gateway of cloud sensor system-based smartliving", Information Fusion (Elsevier). Netherlands, vol. 21, pp.42-56. June 2015.



- [40] Vincenzo Bonifaci, Ho-Leung Chan, Alberto Marchetti-Spaccamela and Nicole Megow. "Algorithms and complexity for periodic real-time scheduling", Journal ACM Transactions on Algorithms (TALG). United Kigdom, vol.9, pp.1-19, November 2012.
- [41] Lu Hou, Shaohang Zhao, XiongXiong, KanZheng, PeriklisChatzimisios, M. Shamim Hossain, and Wei Xiang. "Internet of things Cloud: Architecture and Implementation", IEEE Communications Magazine. Sweden, vol. 54, pp.32-39. June 2016.
- [42] Mirjami Jutila. "An Adaptive Edge Router Enabling Internet of Things", IEEE Internet of Things Journal. Italy, vol.3, pp. 1061-1069, August 2016.
- [43] Chinmaya Kumar Dehury and Prasan Kumar Sahoo. Design and implementation of a novel service management framework for IoT devices in the cloud. The Journal of Systems and Software (Elsevier). Netherlands, vol. 119, pp.149-161 September 2016.
- [44] Chunsheng Zhu, Hai Wang, Xiulong Liu, Lei Shu, Laurance T. Yang, and Victor C. M. Leung. "A novel sensory Data Processing Framework to Integrate Sensor Networks with Mobile Cloud", IEEE Systems Journal. Canada, vol. 10, pp.1125-1136, April 2016.
- [45] Chen Chen, Honghui Zhao, Tie Qiu, Mingcheng Hu, Hui Han, and Zhiyan Ren. "An Efficient power saving polling scheme on the internet of energy", Journal of Network and Computer Applications (Elsevier). Netherlands, vol. 87, pp.58-61, July 2017.
- [46] Liu, R. P., Sutton, G. J., &Collings, I. B. "Power save with offset listen interval for IEEE 802.11 ah smart grid communications", IEEE International Conference on Communications (ICC). Hungry, pp.4488-449 June 2013.
- [47] Poonam Singh, Maitreyee Dutta, and Naveen Aggarwal. "A Review of Task Scheduling based on the meta-heuristics approach in cloud computing", Knowledge and Information Systems (Springer Link). Canada, vol. 52, pp.1-51, September 2017.
- [48] IGI Global. "What is Hybrid Algorithm", <u>https://www.igi-global.com/dictionary/hybrid-algorithm/13449</u>, 2017>[2018-11-22], America, pp.1-9, April 2018.
- [49] Hai-Long Shi, Dong Li, Jie-Fan Qiu, Chen-Da Hou, and Li Cui. A Task Execution Framework for Cloud-Assisted Sensor networks. Journal of Computer Science and Technology (Springer Link). China, vol. 29, pp.216-226., November 2014.
- [50] ShiliangLuo and bin Ren. "The monitoring and managing application of cloud computing based on Internet of Things". Computer Methods and Programs Biomedicine (Elsevier). Germany, vol. 130, pp.154-161. April 2016.
- [51] RadhikaLoomba, Lei Shi, Brenden Jennings, Roy Friedman, John Kennedy, and Joe Butler. "Energy-aware collaborative sensing for multiple applications in mobile cloud computing", Sustainable Computing: Informatics and Systems (Elsevier). Netherlands, vol. 8, pp.47-59, June 2015.
- [52] Min Xia, Teng Li, Yunfei Zhang, and Clarence W. De Silva. "Closed-loop design evolution of engineering system using condition monitoring through the internet of things and cloud computing", Computer Networks (Elsevier), Canada, vol.101, pp.5-118, July 2016.
- [53] Abdulhameed Alelaiwi. "A collaborative resource management for big IoT data processing in Cloud". Clustering Computing (Springer Link). Germany, vol. 20, pp.1791-1799, April 2017.
- [54] Yan, H., Zhang, X., Chen, H., Zhou, Y., Bao, W., and Yang, L. T. "DEED: Dynamic Energy-Efficient Data offloading for IoT applications under unstable channel conditions", Future Generation Computer Systems, Netherlands, vol. 96, pp.425-437, February 2019.
- [55] Mohiuddin, I., &Almogren, A. "Workload-aware VM consolidation method in edge/cloud computing for IoT

applications", Journal of Parallel and Distributed Computing. China, vol. 123, pp.204-214, June 2019.

- [56] Soldatos, J., Kefalakis, N., Hauswirth, M., Serrano, M., Calbimonte, J.-P., Riahi, M. and Žarko, I. P. "Openiot: Opensource internet-of-things in the cloud. In Interoperability and open-source solutions for the internet of things (Springer Link), Ukraine, vol. 4, pp.13-25, December 2015.
- [57] Niranjan K. Ray and Ashok Kumar Turuk. "Routing Protocols in Wireless Sensor Networks", IGI Global Disseminator of Knowledge, America, vol. 4, pp.1-7, November 2016.
- [58] Seokhoon Kim and Wonshik Na. "Safe Data Transmission Architecture Based on Cloud for Internet of Things", Wireless Personal Communications (Springer Link.). Sweden, vol. 86, pp.287-300, March 2016.
- [59] Bhagyashri P. Despande. "The Advanced Way of Data Recovery", International Journal of Computer Science and Applications. India, Vol. 6, pp.241-247, April 2013.



Edje Efetobor Abel is currently a Ph.D. scholar at the School of Computing. Faculty of Engineering, Universiti Teknologi Malaysia. He obtained his MSc (Information Systems Management) in 2010 and BSc (Network Computing) in 2009, in Brunel University, West London, United Kingdom. He is lecturing in Delta State University Abraka Campus, Delta State, Nigeria. His area of interests is Cloud Internet of

Things, Grid Computing, Network Computing, Information Systems Management and Wireless Senor Networks (WSNs).



Muhammad Shafie Bin ABD Latiff received his Ph.D. degree in 2000 from Bradford University, United Kingdom. He is a Professor and currently the Head of Pervasive Computing Research Group and Postgraduate at the School of Computing, Faculty of Engineering, Universiti Teknologi Malaysia (UTM). His research interests are in computer networks with focus generally on routing protocols, grid, and Cloud

computing and wireless sensor networks.